Author's response to reviews

Title: Lifetime total and beverage specific alcohol intake and prostate cancer risk: a case-control study

Authors:

Maddalena Barba (mbarba@buffalo.edu)
Susan E McCann (susan.mccann@roswellpark.org)
Holger J Schunemann (hjs@buffalo.edu)
Saverio Stranges (stranges@buffalo.edu)
Barbara Fuhrman (fuhrman@buffalo.edu)
Sabino DePlacido (sabino.deplacido@unina.it)
Giuseppe Carruba (lucashbl@unipa.it)
Jo L Freudenheim (freud@buffalo.edu)
Maurizio Trevisan (trevisan@buffalo.edu)
Marcia Russell (Russell@prev.org)
Tom Nochajski (thn@buffalo.edu)
Paola Muti (muti@buffalo.edu)

Version: 2 Date: 6 December 2004

Author's response to reviews:

Some changes have been made in accordance to both the requests of the editor in chief, Dr Nehme Gebrayel, and the guidelines of the Nutrition Journal Pre-acceptance list.

I'm describing them by paragraph.

Title page of the manuscript: the only capitol letter that has been used is in the first word of the title.

Authors' affiliations: they are given according to the guidelines. Authors' qualifications are any longer included in the title page.